Author's response to reviews

Title: Swine Influenza and Legionnaire’s disease a Frequently Underestimated Coinfection. A case series

Authors:

Michele Iannuzzi (michele.iannuzzi@unina.it)
Edoardo De Robertis (ederober@unina.it)
Fabio Rispoli (fabio.rispoli@unina.it)
Ornella Piazza (opiazza@unina.it)
Enzo Zamparelli (enzozamparelli@gmail.com)
Giuseppe Servillo (servillo@unina.it)
Rosalba Tufano (servillo@unina.it)

Version: 2 Date: 6 May 2011

Author's response to reviews:

Dear Editor in Chief,

We addressed the referee’s considerations

POINT 1
Many grammatical and typographical errors were corrected. The whole manuscript was extensively edited by mothertongue author.

POINT 2
References were updated according to referee indications

POINT 3
We do agree that buccal swab may not be the ideal test but in non intubated by it’s what the Italian Ministry of Health recommends in Influenza H1N1 pandemics. We do agree with the referee in fact during the prosecution of health care we performed in our patients polymeracse chain reaction on viral RNA from bronco alveolar lavage specimens.

Kind Regards

Michele Iannuzzi